Author's response to reviews

Title: Rationales, design and recruitment of the Taizhou Longitudinal Study

Authors:

Xiaofeng Wang (xiaofengautomatic@gmail.com)
Ming Lu (lu.ming528@gmail.com)
Ji Qian (jijian05@gmail.com)
Yajun Yang (yj229@263.net)
Shilin Li (lishilin0413@gmail.com)
Daru Lu (drlu@fudan.edu.cn)
Shunzhang Yu (szyu@shmu.edu.cn)
Wei Meng (xiaoyanjiangautomatic@gmail.com)
Weimin Ye (weimin.ye@ki.se)
Li Jin (lijin007@gmail.com)

Version: 6 Date: 8 July 2009

Author's response to reviews:

1. Page1 line3-line4: added the affiliation of Ming Lu, Ji Qian, Yajun Yang, Shilin Li, Daru Lu.

2. Page 2, line10: changed 1.6 million to 1.8 million.

3. Figure1, the first box: changed 1.6 million to 1.8 million.